



Article

Disclosure of CSR Performance and Firm Value: New Evidence from South Africa on the Basis of the GRI Guidelines for Sustainability Disclosure

Frank Sampong ¹, Na Song ^{1,*}, Kingsley Osei Boahene ^{1,2} and Kwame Ansong Wadie ¹

- School of Management and Economics, University of Electronic Science and Technology of China, Chengdu 611731, China; fksam29@gmail.com (F.S.); kingohene5@yahoo.com (K.O.B.); kwamewadei@gmail.com (K.A.W.)
- School of Business, Kumasi Technical University, Kumasi 00233, Ghana, West Africa
- * Correspondence: songna@uestc.edu.cn

Received: 25 October 2018; Accepted: 21 November 2018; Published: 30 November 2018



Abstract: Prior CSR and firm performance research has produced mixed results. Even so, numerous researches examining this relationship from the perspective of international standardisation have primarily concentrated on developed economics. This leaves an obvious gap within the extant literature with regards to evidence from sub-Saharan Africa. The aim of this study is to investigate the relationship between the extent of CSR disclosure performance and firm value, in an emerging institutional setting. Using hand collected data of South African listed companies, we apply the GRI G3.1 guidelines, as a measure of disclosure performance. Based on the panel data fixed effect model, we document a positive but insignificant relationship between CSR disclosure performance and firm value. Secondly, a negative and insignificant relationship was found between environmental disclosure performance and firm value. Lastly, we found a positive and statistically significant relationship between social disclosure performance and firm value. Overall, our findings suggest that CSR disclosure has a limited effect on firm value. Our findings hold for a set of robustness tests. Our findings suggest that the incorporation of sustainability disclosure, on the basis of GRI, is moderately high among the selected companies. Implications of our results suggest that CSR disclosure may not necessarily influence firm value, despite its numerous benefits. We contribute to this line of research from a multi-theoretical perspective.

Keywords: corporate social responsibility; firm value; GRI; stakeholder theory; legitimacy theory; South Africa

1. Introduction

The concept of Corporate Social Responsibility (herein termed as CSR) and sustainable development is dynamic, multifaceted, and global, and has proven to be a contentious issue across the globe. For instance, it is assumed that CSR is largely a Western phenomenon, and that it is most likely to be found in countries with globally active organisations, democratic political structures, and active civil society organizations [1]. CSR as a concept has attracted worldwide attention and thus gained global recognition and prominence. The global corporate environment in recent times faces increasing pressure to take up initiatives and recognise its social and environment responsibilities. This has led to a growing interest in the question of whether the adoption of CSR disclosure and strategies by firms that actively support sustainable economic, social, and environmental development can play a key role in the generation of competitive advantage and firm value [2,3].

The notion that businesses have a responsibility of maintaining an impartial balanced working relationship among various stakeholders has been relevant to an understanding of a business strategy

Sustainability **2018**, 10, 4518 2 of 28

for the past half a century [4]. Performance of CSR and disclosure has therefore become an integral part of the corporate fabric [5]. Organisations disclose their social and environmental performance in many different ways, including annual reports, independent reports, websites and press releases, extra supporting documents for annual reports, advertisements, published articles and booklets regarding corporate social and environmental activities, and video tapes [6].

The rise of CSR and sustainability issues globally has led to a planetary rise in global standards for information disclosure over the past decades. Thus, there has been a worldwide diffusion of disclosure standards [7]. A study by Ligteringen and Zadek [8] identified over 300 global corporate standards that have evolved since the 1990s. These standards include the United Nations (UN) Development Goals and Global Compact, the directory of Organization for Economic Cooperation and Development (OECD) for multinational corporations, ISO 14001, the Global Reporting Initiative (GRI) guidelines, Sullivan Global Principles, SA 8000, Series AA1000, and ISO 26000 [7]. These standards encourage disclosures of social and environmental issues of corporate sustainability, among others. As a result of the global rise in standards, there have been calls for enhanced CSR and sustainability disclosures as well as any other information that can potentially impact on organisational performance. Consequently, there has been a rise in the number of organisations disclosing their social and environmental initiatives [9]. According to KPMG reports (KPMG, 2011; KPMG, 2013) 95% of the 250 largest companies in the world (based on the Fortune Global 500 ranking) produced a CSR report in 2011, a 14% increase from 2008. Also, a variety of companies appear to be increasingly engage in CSR disclosure related to environmental and social issues [10,11].

In this paper, we provide evidence in line with recent global upsurge of standards on debates within the broader framework of corporate sustainable development, and the social and environmental disclosure literature of CSR. We highlight the importance of expanding existing evidence on the relations among CSR disclosures and firm performance to different institutional settings that may allow for further findings in light of available inconclusive results within the CSR–FP nexus. The aim of this study is to investigate the relationship between the extent of CSR disclosure performance (and its components) on firm value in an emerging institutional setting where investor protection is relatively low, as a result of a weaker external corporate regulatory environment [12,13]. This study focus on South Africa, with the application of an internationally accepted standard, the Global Reporting Initiative (herein termed as GRI) 2011 (G3.1) guidelines related to sustainability disclosure, as a measure of CSR disclosure performance of sampled companies from the Johannesburg Stock Exchange (herein termed as JSE), an emerging market in sub-Saharan Africa.

Globally, recent trends indicate that many more organisations are paying increasing attention to CSR disclosure, its related organisational performances, and the issuance of CSR/sustainability reports based on international standards. This has become an important issue for both practitioners and academics, as a result of changing business dynamics and importance of organisations in today's global environment, that is, the effects of globalisation and global capitalism [14,15]. As CSR and sustainability reporting are becoming effectively mandatory for global large multinational corporations (MNCs) it has also attracted a considerable amount of academic literature [16,17]. Various stakeholders continue to pressure organisations (i.e., both local and multinational) to improve upon their CSR performances and release the related reports. Such pressures include for instance, public concerns, media interest and advocacy, institutional and regulatory forces, consumer and industrial pressures, reputational concerns, and other perceived market advantages [18]. On the other hand, organisations may disclose their CSR as a way of sustainability development investment, and thus, an evaluation of their consequences should be relevant. Sustainability reporting guidelines developed by the Global Reporting Initiative (herein termed as GRI) provide a more organised approach for companies to disclose their performance on CSR dimensions of sustainability. The reason for the enormity of the subject is that reporting of a firm's actions and inactions has become much more central because investors and other stakeholders are demanding greater transparency about business practices. To this end, it has been asserted that CSR disclosure is now a necessity for all kinds of organisations and is no

Sustainability **2018**, 10, 4518 3 of 28

longer a purview of only large corporations [19]. We argue that the extent of CSR disclosure and its associated transparency should have an impact on the firms' wellbeing.

South Africa makes an interesting study case from sub-Saharan Africa, due to prior research and developments within its context. South Africa is noted for its long history of racial segregation termed as "apartheid". Within apartheid, humanity was largely defined on the basis of race. It is well documented in literature that 'no study of South Africa would be possible without an examination of the regime's racial classifications' [20]. Therefore, it has been argued that the practice of CSR in the context of South Africa was largely "apartheid-induced" [21]. The post-apartheid era therefore saw the formation of affirmative actions, policies, initiatives, and structures by the ANC government (i.e., ANC—the African National Congress is the Republic of South Africa's governing political party and it has been the ruling party since the fall of apartheid in South Africa) after it took over power from the apartheid regime as a democratically elected government. One of the aims of the ANC government on assuming power was to rebuild the corporate image of South Africa, which occurred as a result of the racial imbalances that may have been created after the fall of apartheid. One of the most critical structures that have shaped the corporate and economic landscape of South Africa was the coming into force of the South African Corporate Governance Code of practice (popularly known as the King Reports). The first report was produced in 1994 (i.e., King I), the second in 2002 (i.e., King II), the third in 2009 (i.e., King III), and the most recent was produced in 2016 (i.e., King IV). It has been noted that these reports generally adopt the Anglo-Saxon style of corporate governance [12]; nevertheless, they also included some specific affirmative actions, rules, initiatives and stakeholder provisions.

Firstly, although empirical research on the disclosure of CSR and FP relations has attracted much interest among policy-makers and researchers in the emerging markets context, even so, this relationship seems to be erroneous and to some extent ambiguous. Hence, this relationship remains inconclusive [22]. As prior research in emerging markets on social and environmental disclosure grows in prominence relative to developed markets, a notable avenue by which emerging markets could demonstrate commitments to the global upsurge of standardisation in CSR disclosure trends and sustainable development is to promote a voluntary (and/or mandatory) use of such global standards among local and multinational corporations operating within their markets. It has been asserted that when organisations integrate, adopt, and disclosure on sustainability issues into their operations in a triple bottom context, it is termed as CSR [23]. However, social and environmental research has largely focus on the developed and industrialised economies (i.e., where better institutional settings, higher investor protection, relatively better levels of economic development, and sophisticated financial markets are in existence), with relatively limited prior research in emerging economies [24,25]. There is therefore a lingering need for improve research in social and environmental disclosure in emerging markets to commensurate the global call for corporate disclosures, in line with international standards.

Secondly, the call for CSR disclosures in line with international standards becomes even more desirable as issues of CSR and sustainability are well grounded in the Corporate Governance Codes of South Africa, making this research study unique from the context of South Africa, and for that matter, sub-Saharan Africa. The South Africa Corporate Governance Codes, since its inception, encourages corporations to integrate sustainability concerns into their operations and to disclosure based on the GRI. The King I and King II Corporate Governance Codes required listed companies on the JSE to report based on the GRI guidelines. Mitchell and Hill [26] assert that the GRI reporting in South Africa is endorsed by the King Report for Corporate Governance and the Johannesburg Stock Exchange for listed companies. However, the focal point of this study—the King III Corporate Governance Code—went further to encourage all entities in South Africa to integrate sustainability concerns into their operations, and to report based on the GRI guidelines in a triple bottom context of sustainable development. As such, all entities in South Africa are encouraged to integrate sustainability concerns into their operations, irrespective of the company size, nature, or type—see King III Corporate Governance of South Africa [27]. Therefore, it was timely to undertake research utilising internationally accepted standards (i.e., the GRI guidelines) to investigate the extent of CSR disclosure and its influence

Sustainability **2018**, 10, 4518 4 of 28

on firm value in sub-Saharan Africa, where research is relatively limited. Although literature notes that research on CSR (and disclosure) in emerging markets is relatively limited compared with advanced economies, it is worth recognising the rise in prominence [28] of CSR research in developing countries, and in particular South Africa [29–32]. However, and to the best of our knowledge, no research in today's modern era in the South African context has utilised the GRI 2011 (G3.1) guidelines to investigate the relations among the extent of CSR disclosure performance (including components of CSR disclosure) of public-companies on firm value. Studies on CSR (and disclosure) in South Africa context have largely used affirmative action Acts [31] (for example, the Black Economic Empowerment Act—BEE Act and Broad-base Black Economic Empowerment Act—BBBEE Act), organisational survey data [29] (example, survey data by KPMG International) and other national level instruments [30] (example, JSE Socially Responsible Investment Index) to measure the extent of CSR disclosure. Also, we recognise that the past decade has witnessed research work exploring the usage of the GRI framework and GRI-guided reports.

Consequently, this study makes unique contribution to literature from South Africa context. To the best of our knowledge, this study is different from a study by Ntim [33], which examined the social responsibility role of public companies' contribution towards reducing the negative effects of HIV/AIDS within the realms of social and environmental accounting (SEA), by adopting the GRI 2002 reporting guidelines specifically on HIV/AIDS and how the companies disclose their activities aimed at addressing the menace. Even so, this was a cross-country study within sub-Saharan Africa. Again, this present study is very different from a recent study by Horn et al. [29], which utilises survey data from KPMG International to investigate CSR disclosure in South Africa. Hence, this present study has a different scope which largely differs from the research works by Ntim [33] and Horn et al. [29]. This present study utilises a more recent data and the GRI 2011 (G3.1) of sustainability disclosure, as a measure to investigate the influence of the extent of CSR disclosure performance (and its components) on firm value. In particular, this present study manually collects CSR information disclosed from annual and CSR/sustainability reports of sampled companies to construct three separate disclosure indexes to examine their relations with firm value. In this case, this present study is very unique and different from other research works in the South African context, and as such, makes very substantial contribution to CSR literature in South Africa and in the context of emerging markets.

In this paper, we focus on listed companies on the JSE, South Africa for the following reasons. The rich corporate governance regime has, to a larger extent, shaped the corporate environment of South Africa towards sustainability reporting. In particular, the King III Code on Corporate Governance [27] explicitly encourages all entities in South Africa to integrate economic, social, and environmental issues of sustainability into their day to day operations, and to report based on the GRI. Lastly, South Africa is noted as one of the most ethnically diverse economies in the world [34]. For instance, in 2011, South Africa's population stood at 79% Black Africans, 8.9% were coloured, 2.5% were of Indian/Asian origin, and 9.5% were White (Statistics South Africa, 2012) [35]; see also Gyapong et al. [36] p. 374 (to the best of our knowledge these are the most recent census statistics). This diverse nature makes it unique from its peers in the sub-Saharan region, accompanied with its continuous stability and economic landscape makes it a welcoming destination for a variety of multinational corporations (MNCs). The result of this is the significant rise in the number of company listings on the JSE, which has led to the attraction of more foreign direct investments into the country. Furthermore, South Africa has by far the largest financial system, relatively developed, and most sophisticated in Africa, and it compares well with the financial systems of the developed world, and has secured itself a spot among the top-20 stock markets in the world [37]. The JSE has remained a member of the Federation of International Stock Exchange since 1963, and is the only stock exchange in Southern Africa, but it has gained a mark as one of the largest in the world in terms of market capitalization [37]. With these dazzling features, South Africa compares well with some developed and middle-income emerging economies, hence exploring the extent of CSR disclosure performance of public companies on the JSE from an internationally accepted standard was not far from reach.

Sustainability **2018**, 10, 4518 5 of 28

Using a hand collected dataset of 126 listed South African companies over the period from 2010 to 2015 we investigated the relationship between disclosure of CSR performance, based on G3.1 guidelines related to social and environmental performance indicators, and firm value. To do this, we employed a unique disclosure formula to construct three disclosure indexes. This study finds mixed and interesting results which are consistent with findings of prior studies. First, we examine the influence of the extent of CSR disclosure performance (as represented by social and environmental performance indicators of GRI—G3.1) on firm value. We find that CSR disclosure performance has a positive but insignificant influence on firm value. Even so, and with regards to the correlation matrix below, it also indicates a weak negative relation between CSR disclosure performance and firm value. Second, we examine the influence of the extent of environmental disclosure performance (as represented by environmental performance indicators of GRI—G3.1) on firm value. We find that environmental disclosure performance has a negative and an insignificant influence on firm value. Lastly, we investigated the influence of the extent of social disclosure performance (as represented by social performance indicators of GRI—G3.1) on firm value. We found a positive and statistically significant influence of social disclosure performance on firm value. Overall, our findings suggest that CSR disclosure has a limited effect on firm value. However, the influence of environmental disclosure on firm value may be weakening, if not becoming less important in driving firm performance. Our findings are robust after controlling for the potential effects of endogeneity. Our findings have significant implications in the context of an emerging market with weak investor protection. This study provides support for the theoretical reasoning in line with the stakeholder and legitimacy theories. For instance, the Stakeholder theory posits that a firm's long-term existence depends upon addressing stakeholder concerns, and that CSR has thus expanded the firm's responsibilities to the wider public. CSR disclosures are therefore ways of providing vital non-financial information of stakeholder needs about the firm, which could influence stakeholders' perceptions about the firm and their investment pattern. Also, our findings support firms' legitimacy, that is, where firms use CSR disclosure as a means of legitimising their actions and inactions. Legitimacy theory perceives CSR information as a strategy for managing perceptions of society regarding social and environmental impact on business operations [38]. This theory holds a widespread view that the actions of an entity are desirable, proper, or appropriate within some socially constructed system of norms, values, and beliefs [39]. Therefore the disclosure of CSR has become an integral part of corporate fabric [5].

This present study contributes to the CSR literature in developing countries, in several ways. Firstly, we consider that previous findings on the value relevance of CSR disclosure on firm performance in the developed/industrialised economies cannot be generalised to the emerging markets context, particularly sub-Saharan Africa. Even so, emerging economies, and for that matter sub-Saharan African countries, exhibit wider differences in development relative to other economies. Again, many local and multinational corporations can be perceived differently in the performance of CSR activities. Additionally, stakeholder demands for social and environmental disclosure may substantially differ within different institutional settings. Secondly, although CSR disclosure research has grown in prominence in South Africa, and that, corporate sustainability issues are well incorporated in Corporate Governance Codes of South Africa, CSR disclosure research utilising internationally accepted standards in a triple bottom context was still relatively limited. Therefore, this study shed light on a growing body of literature on international standardisation [40-42] of CSR disclosure towards sustainable development, by utilising the GRI guidelines. Third, and in support of the two points above, our study aims to broaden our understanding and fill an existing gap in literature, by providing new evidence, on the influence of CSR disclosure on firm performance, in line with global debates within the social and environmental disclosure literature, by providing new empirical evidence from a sub-Saharan African context, other than the developed and industrialised markets. Finally, this study makes theoretical contributions, in support of existing literature, that argue for a multi-theoretical perspective to explore the CSR disclosure practice of firms.

Sustainability **2018**, 10, 4518 6 of 28

Our findings have important regulatory, policy, and research implications. This study suggests that CSR disclosure, on the basis of GRI guidelines, is moderately high for our sampled JSE companies over the period of study. This observation holds that CSR disclosure towards sustainable business operations was progressively being upheld in the South African context, and that, efforts by key stakeholders, like the JSE and the South African Corporate Governance Institute, to improve the adoption and incorporation of sustainable development have had positive impacts on companies' operations. However, and like most administrative structures which are not without caveats; we suggest that attention be given to the credibility of some information that are disclosed by companies. From our findings we argue that the influence of CSR disclosure performance, on the basis of the GRI guidelines, serves to provide a common playing field to enable companies to be assessed from an internationally accepted standard, which is vital for performance evaluation.

The rest of this paper is organised as follows. Section 2 provides theoretical background, prior literature, and hypotheses development. Section 3 introduces the data, research design, and methodology. Section 4 presents the analyses and discusses the empirical results. Section 5 presents the robustness test. Finally, Section 6 presents a summary and conclusion.

2. Theoretical Background, Prior Research and Hypotheses Development

2.1. Multi-Theoretical View

Many different theories have been identified to explain CSR disclosure patterns, its importance, and necessity [43]. These theories and approaches are under the themes of economics, politics, social integration, and ethics [44,45]. Recognising that there may be limitations with respect to using a single theoretical perspective to explore the notion of CSR disclosure, as well as, given the multiplicity within corporate motivations for CSR disclosures [46], this present study adopts a multi-theoretical view to investigate the relationship between the extent of CSR disclosure performance and firm value of public companies on the JSE. Consistent with prior suggestions and evidence [31,47,48], we consider that these theories are complementary rather than opposing. Therefore, we propose that a combination of the theoretical perspectives is useful to explore CSR disclosure practices. Hence, this present study employs the Stakeholder theory and Legitimacy theory. Even so, it has been asserted that the Stakeholder theory is closely aligned with Legitimacy theory and that the two are often used to complement each other [49]. Again, recognising the existence of extensive literature on these theories, which is well documented in literature, we briefly describe these theories.

Stakeholder theory posits that a firm's long-term existence depends upon addressing stakeholder concerns, and that CSR has expanded the firm's responsibilities to stakeholders and to the wider public [50,51]. Porter & Kramer [52] document that CSR can be much more than a cost, constraint, or charitable deed and that it can, in fact, be a potent source of innovation and competitive advantage. Being socially responsible brings firms various potential benefits such as improved labour relations and employee productivity, less risk of litigation related to its products/services, fewer complaints from the community because of environmental issues, lower regulatory costs, and better brand recognition, among others. These potential benefits suggest that socially responsible firms will perform better. Stakeholder Theory [50], therefore, posits that corporate disclosure is an instrument for managing the informational needs of various powerful stakeholder groups. Theoretical literature has formulated competing hypotheses on the effect of CSR on firm performance. The first one is called "social impact hypothesis" and the second is the "shift of focus view". While the former claims that CSR has a positive effect on firm performance, the latter suggests the opposite. According to the social impact hypothesis, three channels may explain the positive relationship between CSR and firm performance. These are improving employee productivity by providing a better working environment [53]; increasing managerial know-how and thus enhancing organizational efficiency [54]; and increasing social reputation, trust [55], brand image, and product competitiveness [56]. Lastly, modern Stakeholder theory contends that the value of the firm depends on the cost, not only of explicit Sustainability **2018**, 10, 4518 7 of 28

claims, but also of implicit claims. From this viewpoint, the set of claimants on the firm's resources goes beyond the stockholders and bondholders to include stakeholders who have explicit claims on the firm, like wage contracts and others with whom the firm has made implicit contracts, involving, for instance, quality service and social responsibility [57].

Legitimacy theory perceives CSR information as a strategy for managing perceptions of society regarding the social and environmental impact on business operations [38]. Legitimacy theory holds a widespread view that the actions of an entity are desirable, proper, or appropriate within some socially constructed system of norms, values and beliefs [39]. Consequently, stakeholders can influence corporate actions by arguing that a given practice does not conform to societal expectations and/or lacks legitimacy [58,59]. Hence, when the legitimacy of an organisation is endangered, it could employ some legitimisation strategies to minimise those threats by seeking to influence stakeholder actions and perceptions [48,60]. For instance, within the context of South Africa, it has been argued that organisations could legitimise their actions by disclosing BEE information, notwithstanding its usefulness in explaining CSR disclosures motives [31]. De Villiers and Van Staden [38] asserted that in terms of legitimacy, when companies face changes in societal expectations they will change the extent and pattern of their disclosures in order to remain legitimate with their powerful stakeholders. These authors provide justification to support an understanding of legitimacy by adding a dimension of reduced disclosure levels by organisations as a legitimising strategy [38]. Therefore, organisational legitimacy theory predicts that corporations will do whatever they regard as needful in order to preserve their image as a legitimate organisation with legitimate aims and methods of achieving it [38]. Legitimacy has largely been used to support the idea that social disclosures will be maintained at present levels, or increased over time, to avert legitimacy crises. Proponents of legitimacy [38] have therefore predicted that maintenance and reductions, as much as increases, in social and environmental disclosure levels of firms are influenced by way of legitimacy. Woodward et al. [61] have shown that both stakeholder theory and legitimacy theory consider an organisation to be part of the wider social system. This notion is consistent with foundation pillars of the King III Corporate Governance Code. Whilst legitimacy theory looks at society as a whole, stakeholder theory recognises that some groups within the society are more powerful than others. Hence, disclosures are largely driven by stakeholder and/or societal pressure, and that such disclosures are likely to reduce information asymmetry.

2.2. GRI Guidelines and CSR Disclosure

The GRI guidelines are the first global standards for sustainability reporting. They feature a modular, interrelated structure and represent the global best practice for reporting on a range of economic, environmental, and social issues. The GRI reporting guidelines on CSR information has widely been acknowledged as a leader in the international standardisation of sustainability disclosure [62]. Further, it is regarded a primary tool for CSR disclosure, as it has worldwide applicability in multinational organisations that operate in a variety of industries [63]. It has been asserted that, firms that incorporate the GRI guidelines into their operations appear to have higher levels of commitment to CSR than those firms that do not incorporate these standards. Even so, opponents of the GRI framework have criticised it from a theoretical viewpoint that its sustainability principles seem to be widely dispersed [63]. Furthermore, the lack of specific, formal national and international regulations seems to allow organisations much flexibility on how they carry out their social and environmental disclosure activities, while allowing the biased use of these guidelines [64]. A research that used GRI guidelines as CSR reporting templates claimed that companies may increase their symbolic performance by being able to "tick more boxes", thereby increasing the measured volume of their reporting framework. However, these authors note that such greater measured volume may not necessarily indicate managers' intention with regards to social and environmental disclosure issues [65]. Nevertheless, supporters of social reporting and the GRI have long contended that disclosure offers reporting companies a wide spectrum of intangible benefits, such as employee loyalty and consumer reputation. Also, new research suggests that the value of CSR disclosure extends

Sustainability **2018**, 10, 4518 8 of 28

to the firm's balance sheet [66,67]. An analysis of the results of more than 200 independent empirical studies examining the relations of corporate social and environmental performance to corporate financial performance suggested that companies might benefit from increased communication of their good deeds [68]. The studies in the sample specifically covering transparency and reporting indicated positive market reactions to sustainability reporting.

2.3. The Questionable CSR-Firm Performance Relationship

The debates on the relations between CSR and firm performance (herein termed as FP) have been part of the academic literature for decades, and therefore are not new [69]. Reviews of the some relevant studies on this questionable relationship have shown both positive, negative, and to some extent no relations at all [53,54,70–73]. Such inconclusive findings create a research gap for further examinations in an attempt at reaching conclusive results. The literature has attributed the differences in results to research design adopted which according to scholars may likely introduce error in specifying the models or issues of endogeneity. Usually, competition between the available and research adopted has also led to results that are incompatible and has helped perpetuate conceptual inconsistencies and contradictory arguments with the CSR and FP nexus [53,54]. The existing literature indicates that researches investigating the relationship among variables are associated with the endogeneity problem. This present study claims that causality and hence the problem of endogeneity exists. Therefore, we briefly explore the potential problem of endogeneity. Prior studies have examined the potential effects of endogeneity and their remedies [74]. Li [74] provides evidence by examining a variety of econometric methods that researchers commonly use to address endogeneity in empirical corporate finance. According to Li [74] the problem of endogeneity is one single biggest issue in the research line exploring the relations among variables of interest in the corporate finance literature, which is the main reason why results are largely mixed and inconclusive. The author asserts that the endogeneity problem has always been the main or possibly only obstacle to understanding the true relations between different aspects of constructs in empirical corporate finance. He observed further that variables are typically endogenous, instruments are scarce, and causality relations are complicated. Even so, researchers either use one or two simple methods to mitigate the issues of endogeneity or simply ignore it totally. Li [74] illustrates this by experimenting with a variety of econometric methods on how to mitigate the endogeneity problem. The author shows that all the prevailing econometric methods' remedies are generally effective in mitigating the potential problem of endogeneity to some degree and to correct the sign from positive to negative. However, the effects vary considerably in quantitative terms. Among the remedies identified, Li [74] finds the generalised moments model (GMM) to having the greatest correction effect on the bias, followed by instrumental variables, fixed effect models, lagged dependent variables, and the addition of many more control variables. Even so, Li [74] argues that a combination of the methods, like firm fixed effects, year fixed effects, and the addition of more meaningful control variables appear to work well, even without a valid instrumental variable. As we will see in subsequent sections of this paper, we deal with this issue by employing several methods in an attempt to mitigate the potential effects of endogeneity in this present study.

Literature has further attributed the differences in results to the non-availability of a uniform definition on CSR. Thus, different scholars and researchers have used numerous different CSR definitions between different studies and settings to comprehend their research's idea. Also, researchers contend that CSR disclosure and performance may vary across different institutional and industrial setting, because of the various costs and benefits associated with diverse institutional and industry characteristics [75,76]. For instance, it has been shown that industry sensitivity to the environment will affect disclosure responsibility of firms [77]. The use of different definitions for CSR by researchers, between different studies and settings, has been done to comprehend the researchers' idea clearly, although there seems to be substantial common ground between them [78]. Therefore, this present study follows an acceptable concept of sustainable development as coined by the Brundtland Report, which claims for a development that "meets the needs of the present without compromising the ability

Sustainability **2018**, 10, 4518 9 of 28

of future generations to meet their own needs" [79] (p. 8). The report defines corporate sustainability under three umbrellas: economic attainment, social justice, and environmental stability [79]. As a result, CSR has now been viewed to achieve balance between the three dimensions; on an economic benefit level—businesses should be profitable and supply the society with products and services; on the ecological level—businesses must protect the environment and find new ways leading to eco-efficiency; and on the social level—businesses should safeguard human rights and promote social integration [80]. Werther and Chandler [81] state that the concept of CSR is about integrating the triple bottom line of social, economic, and environmental dimensions in a multi-stakeholder dialogue on a voluntary basis.

2.4. Prior Research on CSR Disclosure Performance and Firm Value

To date, the existing empirical studies on the CSR-FP relations in both advanced and emerging economies report mixed results. In contrast, it has also been found that CSR reputation index positively affects firms' financial performance [82,83]. Furthermore, Orlitzky et al. [54] conducted a meta-analysis of 52 studies and showed that social responsibility disclosure and, to a lesser extent, environmental responsibility disclosure is likely to have a positive effect on firm performance. Theoretically, the formation and evolution of the Stakeholder Theory [84], and its instrumental aspects in particular [85], has provided the foundation for proponents of CSR to theorize that CSR can help in constructing and concretizing relations of trust with a variety of stakeholders (i.e., employees, consumers, local communities, environmental activists, and concerned citizens among many others) that are important to the long term success and value of a firm. Several researchers have for that matter documented a positive association between CSR-FP relationships [83,86]. On the other hand, there are opponents to this notion. Historically, Milton Friedman [87], an eminent economics professor, has been one of the most outspoken opponents of CSR. In his famous article entitled 'The social responsibility of business is to increase its profits', published in 1970, he argued that the application of CSR imposes unjustifiable and fundamentally undemocratic tax on shareholders, that its implementation cost outweighs any potential tangible benefits, and as such, it constitutes misappropriation and allocation of valuable resources of the firm [87]. These two opposing views (including those of other scholars that fall within the spectrum as defined), have led to the empirical investigations from different angles regarding the extent to which CSR can lead to demonstrably superior firm performance or not, by academic researchers. Therefore, with the recent increase in global attention on social and environmental issues toward sustainable development, this has led to an equal rise in global standardisation on social and environmental disclosures, towards the creation of a shared future. The rise in global standardisation has further provided academia with an even greater avenue to investigate deeper into the alleged relations between CSR disclosure performance and firm performance. Researchers have noted that the term "association" as used in the CSR and FP relations is a broad term, which is simultaneously used to represent the influence of independent variables on dependent variables, as well as, the relationships among them [69,88]. Therefore, the past several years witnessed scholars and researchers making use of several conceptual and theoretical foundations to develop and test hypotheses regarding the nature of the CSR-FP relations. Hence, results on the CSR-FP relations have been mixed; depending on theoretical orientations.

In support of a neutral CSR–FP relation, McWilliams and Siegel [89] argue that in equilibrium there should be no significant CSR–FP relations. Using the supply and demand theory of the firm, these authors [89] contend that managers' main objective is to maximize shareholder wealth. Accordingly, managers will choose the level of the activities (including CSR activities) that maximize firm value, given the demand for various characteristics and the costs of supplying them. Thus, in equilibrium, there is a neutral CSR–FP relation. Linked to this neutral relation, is the argument that the environments in which societies and organisations exist are too complex that a simple and direct CSR–FP relationship seems questionable [82]. One of the bases for predicting a negative CSR–FP relationship derives from the neoclassical theory of the firm. This theoretical view posits that the opportunity cost of expenditures

Sustainability **2018**, 10, 4518 10 of 28

for corporate responsibility disclosure performance exceeds the profitability of such investment, so that a trade-off exists between CSR-FP relations. Therefore, where stakeholders exert effective pressure for CSR performance, the outcome is an observed diminished firm financial performance and firm value. Scholars [90] argue that, in spite of the trade-off between CSR and FP, managers may undertake CSR activities for their own benefit (e.g., public fame) at the expense of shareholders value maximisation. These scholars [90] argued that managers may seek to divert stakeholders' attention of the poor firm financial performance by promoting the firms' CSR activities. In contrast, managers may reduce expenditures on CSR programs in order to boost short-term profitability to enhance their personal compensation. Either way, CSR may signal to investors that managers are liable to acting in their own benefit. In this way, it is expected that stock prices will fall as investors come in the known that managers of socially responsible firms act in ways detrimental to shareholder interest [69].

Although researchers have observed mixed results in the literature regarding the CSR–FP relationship, prior evidence largely points towards a positive relationship in majority of the studies conducted in the developed countries context. However, and relating to developing economies, the evidence is still mixed and inconclusive. Some researchers found CSR to be positively and significantly associated with firm value/performance, but a large number of studies revealed an insignificant association. The difference in results, as noted earlier, may be due to differences in methodology adopted, study period examined and the wide variation in variables used in measuring CSR, the different institutional setting, and to some extent the operational constructs [91], and importantly the problem of endogeneity [74]. According to Carrots and Sticks [92], there exists a positive theoretical relationship between CSR disclosure and firm value/performance. Therefore, from the evidence above, we formulate the following first alternative hypothesis for empirical testing in the context of South Africa.

Hypothesis 1 (H1). *Ceteris paribus, there is a positive and significant relationship between the extent of CSR disclosure performance and firm value of selected JSE companies.*

Regarding the relationship among the components of CSR disclosure performance (i.e., environmental disclosure performance and social disclosure performance) and firm value/performance, the results are also mixed and inconclusive. Although general and theoretical notions point towards a positive relationship, other researchers have found a negative relationship and, in some cases, no relationship at all. In all the different findings, researchers have provided theoretical basis to justify their results. In this present study therefore, we propose that the components of CSR disclosure performance (i.e., social and environmental disclosure performance) will have a positive relationship with firm value/performance, and hence, based on the above theoretical foundations and prior evidence, the following second and third alternative hypotheses are formulated for testing in the context of South Africa:

Hypothesis 2 (H2). Ceteris paribus, there is a positive and significant relationship between the extent of environmental disclosure performance and firm value of selected JSE companies.

Hypothesis 3 (H3). Ceteris paribus, there is a positive and significant relationship between the extent of social disclosure performance and firm value of selected JSE companies.

3. Data, Research Design and Methodology

3.1. Data: Sample Selection, Source and Description

The present study focus on companies listed on the JSE. Data were hand collected from annual reports, which were obtained from the African Financial Market database. In addition, we used CSR and sustainability reports which were collected from websites of sampled companies, for the period

Sustainability **2018**, 10, 4518 11 of 28

of 2010 to 2015. Consistent with prior disclosure studies, our final sample of 126 listed companies, is made up of only nonfinancial companies. In arriving at our final sample, we excluded companies with two or more missing years' annual and CSR/sustainability reports. Our sample is made up of data over a six-year period for an unbalanced panel data analysis to explore the relationship between the extent of CSR disclosure performance and firm value. Therefore, our final sample of 126 listed companies, over a 6-year period which produced a total of 747 observations, within seven industry sectors. Researchers have noted that sample size is an essential piece for any empirical study. Thus, the sample size is supposed to represent a bigger population that would be too hard to measure. Where the sample is too small it may lose its significance, which means that the results may not be extendable to a larger population since the chosen sample was not largely representative [93]. Hence, and to the best of our knowledge, our final sample size of 126 listed companies over a six-year period, for this present study, provided a larger observation compared with prior CSR disclosure research work [29,31] in the context of South Africa.

3.2. Measurement of Variables and Model Specification

Our main independent variable is CSR disclosure performance. However, for this present study, we construct three disclosure indexes (i.e., CSR disclosure performance index, social disclosure performance index, and environmental disclosure performance index). In constructing our disclosure indexes, we manually extracted the CSR information disclosed (i.e., social and environmental disclosure information) from the annual reports and CSR/sustainability reports of sampled companies. We then analysed the CSR information collected, in line with G3.1 guidelines, to ascertain the extent of CSR disclosure performance. Thus, we used the G3.1 guidelines of 75 comprehensive provisions (i.e., measured on the basis of environmental and social performance indicators) as a measure of extent of disclosure performance. Binary coding using content analysis technique was employed for this case. The CSR disclosure performance index is developed based on 75 provisions of G3.1; the social disclosure performance index is developed based on 45 provisions of G3.1 and the environment disclosure performance index is developed based on 30 provisions of G3.1 guidelines. Thus, the CSR disclosure performance index is a combination of the social and environment performance indicators. Supplementary Material shows details of the G3.1 guidelines as provided by the GRI. Hence, consistent with prior disclosure research [36,94], we award a company 1 if a specific CSR item disclosed is in line with the G3.1 guidelines (i.e., either full or partially disclosed); otherwise 0. However, and consistent with prior research [95] where a performance indicator is found not to be applicable to a particular industry, we do not penalise nor award a score. Thus, using binary coding system of content analysis, we score the sampled companies based on their CSR information disclosed, measured against the G3.1 disclosure performance indicators, over the period of observation. Prior research has extensively used content analysis in disclosure research. Hence, this study used binary coding system of content analysis.

This present study employs the GRI (G3.1) guidelines because of its comprehensiveness in support of sustainability development and CSR disclosure from an international perspective. Also, it falls within our study period: 2010 to 2015. Our study is further guided by provisions of the South Africa Corporate Governance Codes, in particular, the King III Report, which encourages all entities in South Africa (including listed companies) to incorporate into their operations sustainability issues in a triple bottom context, and to report based on the GRI. The study period also falls within the period of the King III Report, which came into force from 2010 to 2016, following the inception of the King IV Report. We wish to state emphatically that in constructing the CSR disclosure index for this present study, we do not identify companies that issued GRI-guided reports, as have been used by some prior researchers [96–98]. Rather, and unique to this study, we manually review (measure) CSR disclosed information collected from the sampled companies' annual reports and CSR/sustainability reports against the G3.1 guidelines, to ascertain the extent of disclosure with this guideline. For that matter,

Sustainability **2018**, 10, 4518 12 of 28

the authors thoroughly read all annual and CSR/sustainability reports twice to ensure consistency in the scoring.

Tobin's Q is used as a measure of firm value and the dependent variable. This is because it has widely been used in prior research [99,100] as a measure of long term performance. Scholars [101] have noted that firm value incorporates both a firm's future cash flow and the assessed inherent risk reflects in the cost of capital. Thus, firm value highlights the incremental portion of the market price that exceeds the book value. Therefore, Tobin's Q is most suitable for the measure of firm value; even so, this study seeks to examine the influence of CSR disclosure performance on long-term firm value, which makes the Tobin's Q a preferred indicator. Other financial performance variables like return on assets (ROA) and return on equity (ROE) are relatively unsuitable in this case because they are short-term measures of firm performance [102,103]. Also, unlike Tobin's Q, other financial measures of firm value/performance (like ROA and ROE) could be easily subjected to several short-term earnings manipulation activities.

Researchers have noted that several variables could influence both independent and dependent variables. It is therefore not unusual to see important independent variables omitted in regressions for several reasons, such as when variables are immeasurable, the presence of an in-exhaustive list of potential independent variables, when the relationship between the omitted variables and variables that are included cannot be measured, and when the effect of the omitted variables are unknown [104]. Thus, omitted variable bias can cause regression estimates to be inconsistent [105]. Traditionally, omitted variables are dealt with by using either proxies [104]. Therefore, and consistent with prior studies, this present study employs several control variables to minimise potentially omitted variable bias. The following control variables are included in this study, board independence, presence of an external audit firm, firm size, return on asset (representing profitability), firm age, and leverage. For instance, Satish and Thompson [106] included in their study seven control variables. This is because previous studies have indicated that some variables may have the potential to influence both the dependent and independent variables. Therefore, and as argued, the control variables are adopted to account for factors other than the theoretical constructs of interest that could explain variations in the independent and dependent variables.

Empirical research in corporate finance considers firm size an important and fundamental firm characteristic, since they are usually the key variables that affect the independent and dependent variables simultaneously. Hence, the literature asserts that firm size is closely related to CSR and firm performance. Therefore, and importantly, the study by Dang et al. [107] provides empirical evidence for a "measurement effect" in the "size effect". Specifically, these authors examine the influences of employing different proxies of firm size (i.e., total assets, total sales, and market capitalization) in areas of empirical corporate finance research. Dang et al. [107] highlights numerous empirical implications: first, the coefficients of firm size measures are robust in sign and statistical significance. Second, the coefficients on regressors other than firm size often change sign and significance when different size measures are employed. Third, the goodness-of-fit measured by R-squared also varies with different size measures, suggesting that some measures are more relevant than others in different situations. Last, different proxies capture different aspects of "firm size", and thus have different implications. Therefore, the choice of size measures needs both theoretical and empirical justification [107]. Kumar et al. [108] provides a comprehensive review of literature and classified the theories into four categories, these are, technological theories, organizational theories, regulatory theories and financial theories (as cited in Dang et al. [107]). Existing literature therefore show that "firm size" is a most significant variable in most studies in this area. Therefore consistent with prior studies, we use total assets as our choice of firm size. We do this in line with the suggestion by Hart and Oulton [109] that the choice of a measure used depends on data availability (we do so not out of convenience), and also it is in line with the purpose of this study. However, to mitigate for the high skewness associated with the firm size data, we employ the natural logarithm of the total assets value. All variables are listed and defined in Table 1 below. Consistent with prior disclosure research [91]

Sustainability **2018**, 10, 4518 13 of 28

that employed disclosure index, this study adopts the disclosure formula below to construct the three separate disclosure performance indexes.

$$DISC_{ij} = \frac{\sum\limits_{i=0}^{n} X_{ij}}{m_{j}} \tag{1}$$

where m_j = the maximum expected score for each category, j is the company, i = the items and X_{ij} assumes a value of 1 if a company disclosed an item, otherwise 0. However, we do not either penalize or award a company where an indicator is not applicable to a particular industry.

Table 1. Definition of variables.

Dependent Variable						
Q	Total market value of the firm to total asset value					
Independen	t Variables					
Env_DISC Soc_DISC CSR_DISC	Soc_DISC Consisting of 45 disclosure performance provisions of GRI, G3.1					
Control Vari	ables					
FSIZE BIG4AUD FAGE BODIND ROA LEV	Natural logarithm of year-end of total assets A dummy variable equal to '1' if a firm is audited by a Big Four audit firm, otherwise '0' Number of years from a firm's founding year to 2010, start year for the study period Number of independent executives and nonexecutive directors on corporate boards Ratio of operating profit to total assets Ratio between the volume of a firm's short and long term debt to its total assets					

Note: Q = Tobin's Q; Env_DISC = environmental disclosure performance index; Soc_DISC = social disclosure performance index; CSR_DISC = CSR disclosure performance index; FSIZE = Firm size; BIG4AUD = an external big four audit and assurance company i.e., PricewaterhouseCoopers, Deloitte & Touché, Ernst & Young, and KPMG; FAGE = Firm age; BODIND = board independence; ROA = Return on Asset; LEV = leverage.

The issue of reliability and the approval of the quality of content analysis adopted for a particular study are very important. As such, prior studies [110] have observed different ways to improve the quality of content analysis. The process involves initial identification, selection, and definition of the disclosure items from related literature which is of utmost importance. Secondly, it is necessary to establish a reliable coding system with proper decision steps. Finally, the adoption of a uniform coding system is highly essential for a reliable content analysis [111]. Sustainability reporting on the basis of GRI guidelines for disclosure of CSR information has been noted as very reliable for analysing firms' sustainable disclosure performances [7]. Therefore, in order to measure CSR disclosure performance, the G3.1 guidelines which provides a total of 75 well-defined specified performance indicators for all the components of CSR disclosure (i.e., environmental and social disclosure performance) was used, as a base to measure the CSR information disclosure collected from annual and CSR/sustainability reports. Moreover, it has been reported that, according to the GRI, only the validated contents are disclosed. Thus, the present study uses GRI-guided disclosure performance indicators as a unit of measure along with a uniform coding system for enhancing the reliability [91].

3.3. Model Specification

Consistent with prior research [112], and assuming that all relations are linear, the following regression model is employed to test the stated hypotheses and to establish the relationship among the dependent and independent variables.

$$Q_{it} = a_{it} + DISC_{it} + \sum_{i=1}^{n} \beta_1 Controls_{it} + \varepsilon_{it}$$
 (2)

Sustainability **2018**, 10, 4518 14 of 28

where Q refers to Tobin's Q; α refers to constant term; DISC refers to CSR_DISC, Soc_DISC, and Env_DISC indexes and; CONTROLS refers to the control variables of firm size (FSIZE), firm age (FAGE), presence of an external audit firm (BIG4AUD), board independence (BODIND), profitability (ROA), leverage (LEV), and Industry fixed effects.

4. Empirical Results and Discussion

4.1. Descriptive Statistics

Table 2, shows the descriptive statistics for all variables relating to mean, standard deviation, skewness, and kurtosis included in this present study. Here, our aim is to get an idea about how the variables are distributed, to examine the inherent interrelationships among them, and to know if there exists the presence of outliers in the data set.

Variable	Obs	Mean	Std. Dev.	Skewness	Kurtosis
Q	747	2.35	4.39	4.19	23.15
Env_DISC	747	0.61	0.16	-0.12	2.52
Soc_DISC	747	0.66	0.09	0.56	4.07
CSR_DISC	747	0.65	0.10	0.52	4.98
FSIZE	747	22.08	2.52	-0.23	3.49
BIG4AUD	747	0.76	0.43	-1.23	2.51
FAGE	747	36.76	29.64	0.91	2.94
BODIND	747	7.06	2.97	0.08	2.59
ROA	747	0.46	7.27	23.15	577.62
LEV	747	0.52	2.10	26.72	724.82

Table 2. Summary of descriptive statistics.

Authors construct from STATA. Note: All variables are as described and defined in Table 1.

Table 2, provides a summary of all study variables used. The table shows that Q has a mean of 2.35 and a standard deviation of 4.39. The mean score of Q suggests that on average the sampled companies have high market value/performance over the study period. However, the standard deviation for Q shows wider variations within the firm value of the sampled companies. On the other hand, the table also shows that the mean and standard deviations of our main independent variables respectively are: Env_DISC is 0.61 and 0.16; Soc_DISC is 0.66 and 0.09; and CSR_DISC is 0.65 and 0.10. These indicate that there exist a high degree of similarity among them, and that the differences among them are not very wide. The mean score for our main independent variables all indicate an average of 61% and above; this gives an indication that our sampled listed companies, for the period under observation, performed moderately high for CSR disclosure performance, as measured on the basis of the G3.1 guidelines for disclosure of CSR information. Therefore, this suggests that the extent of incorporation of sustainability among ompanies on the JSE is high. However, the presence of the wide variations within the scores suggests that there exist a high degree of heterogeneity among sampled companies. Additionally, the observed values of Skewness for some of the variables and the near equality of the scores of our main independent variables indicate that the distribution is not far away from symmetry.

Table 3 below presents the correlation matrix among all the variables in this study. The correlation matrix provides understanding of the degree of relations among all the variables. First, we look at the relations amongst our variables of interests, the dependent and the independent variables. It shows weak negative relations between Q and Env_DISC of -0.12, Q and Soc_DISC of -0.08, and Q and CSR_DISC of -0.13, respectively. These results imply that CSR disclosures do not necessary lead to increased firm value. Next, we look at the relations between the dependent variable Q and the control variables. It further shows weak negative relations between Q and all the other variables as follows. Q and FSIZE at -0.25, Q and BIG4AUD at -0.10, Q and FAGE at -0.12, Q and BODIND at -0.13, Q and ROA at -0.001, and Q and LEV at -0.02. Relations among independent variables and control variables:

Sustainability **2018**, *10*, 4518

FSIZE and Env_DISC is 0.13, FSIZE and Soc_DISC is 0.23, and FSIZE and Over_DISC is 0.22, all indicate a weak positive relation. FAGE and Env_DISC is 0.04, FAGE and Soc_DISC is 0.19, and FAGE and CSR_DISC is 0.13; all indicate a weak positive relation. From Table 3, we find some high correlation among the variables. These high figures may raise concerns of multi-collinearity [113] within the correlation matrix. Scholars [114] recommend ways of addressing multi-collinearity. First, regression techniques that use biased regressors coefficients can be used to achieve a substantial reduction in the stability and variance of these same coefficients. Second, the theoretical model can be altered either by combining the collinear independent variables or by including them in different regression models. Consistent with prior suggestion [114] that a safer approach to addressing multi-collinearity is to include the suspected variables in alternative regression models, hence, we adopt this approach. Therefore, we are of the view that the high correlation among variables in this study is not an issue. Hence, we run two separate regression equation models for this present study. Regression equation Model 1 is used to investigate the relationship between CSR disclosure performance and firm value. Also, regression equation Model 2 is used to investigate the relationship among social disclosure performance and environmental disclosure performance and firm value. In each of the regression models, we include all the control variables to minimise potentially omitted variable bias.

1 2 3 7 8 9 10 5 6 1.Q 1.000 2.Env_DISC -0.1161.000 3.Soc_DISC -0.0840.279 1.000 4.CSR_DISC -0.1260.837 0.758 1.000 5.FSIZE -0.2460.131 0.230 0.220 1.000 1.000 6.BIG4AUD -0.1010.090 0.114 0.126 0.278 0.228 0.200 1.000 7.FAGE -0.1160.0400.1880.1340.542 0.257 -0.1270.137 0.106 0.336 1.000 8.BODIND 0.0410.059 0.030 0.029 9.ROA -0.0010.017 -0.0000.040 1.000 -0.01410.LEV -0.029-0.039 -0.042 0.007-0.063-0.044-0.039 1.000 -0.018-0.056

Table 3. Correlation matrix.

Authors construct from STATA. Note: All variables are as described and defined in Table 1.

4.2. Panel Data Regression

In this study, both cross-sectional and time series components are involved. Hence, given the nature of data and to allow for time series and cross-sectional data observations, we employed the panel data technique. Prior studies in the context of South Africa have utilized both balanced panel [31] and unbalanced panel [36] data techniques. Therefore, consistent with these authors and based on data collected, this study employs an unbalanced panel data technique. Many different reasons have been associated for the use of unbalanced panels. Most importantly, and applicable to this study, in the real world most panels are unbalanced [115]. Therefore, to choose a balanced panel may introduce sample selection bias into the dataset, which may result in less representation of the population. Consistent with prior studies [116], this study employs two-panel data techniques, which are the fixed effects and the random effects models, to control for possible unobserved firm-level heterogeneities. However, before deciding on an appropriate panel estimation technique, we conduct the Breusch and Pagan test (test results not reported in this study) to determine the suitability of either a pooled OLS or random effects regression model. The test rejects the null hypothesis of zero variance across entities, which makes pooled OLS unsuitable. Afterwards, we conduct the Hausman test to identify if the individual effects are fixed or random, in cases where individual effects exist. Thus, we used the Hausman test to decide on the suitability of either the fixed effects or random-effects model. The result of the Hausman test rejects the null hypothesis that individual effects are uncorrelated with the regressors, Sustainability **2018**, 10, 4518 16 of 28

for regression Models 1 and 2. Hence the fixed effects panel regression model is used for this present study to investigate the relationships among the variables.

To investigate the relationship among CSR disclosure performance (and its components of Social disclosure performance and environmental disclosure performance) and firm value, we ran the fixed effects model. Tables 4 and 5 below present the results of the panel regression using the fixed effects for Regression Model 1. The relevant statistics to take note of in the tables below are the coefficient (β) of the regressors and the *p*-values. The significance level is set to 95% significance, with *p*-values at 1, 5, and 10% interpreted to be statistically significant.

Q	Coef. Std. Err. t $p > t $ [95% Conf. Inte			. Interval]		
CSR_DISC	0.313	1.130	0.28	0.782	-1.906	2.532
FSIZE	-1.232	0.227	-5.44	0.000	-1.677	-0.787
FAGE	0.234	0.044	5.34	0.000	0.148	0.320
BIG4AUD		0 (omitted)				
BODIND	0.169	0.048	3.52	0.000	0.075	0.263
ROA	0.004	0.009	0.39	0.697	-0.015	0.022
LEV	0.012	0.035	0.35	0.729	-0.056	0.080
_cons	19.563	4.311	4.54	0.000	11.097	28.028
Industry-effects	YES					
Sigma_u	8.165					
Sigma_e				690		
Rho	0.959 (fraction of variance due to u_i)					
F test that all u_i=0:	F(125, 615) = 32.59 $Prob > F = 0.0000$				100	

Table 4. Fixed Effects Model 1 (with CSR disclosure performance).

Note: sigma_u = SD of residuals within groups ui; sigma_e = SD of residuals (overall error term) ei; Rho = 95.9% of the variance is due to differences across panels; 'rho' is known as the intra-class correlation; t = t-values test the hypothesis that each coefficient is different from 0. To reject this, the t-value has to be higher than 1.96 (for a 95% confidence). If this is the case then you can say that the variable has a significant influence on your dependent variable (y). The higher the t-value the higher the relevance of the variable; p > |t| = two-tail p-values test the hypothesis that each coefficient is different from 0. To accept this, the p-value has to be \leq 0.001, \leq 0.05, and \leq 0.10 (at 95% confidence level); if this is the case then you can say that the variable has a significant influence on your dependent variable (y). Source: Authors' result of fixed effects model from STATA.

Fixed Effects (within) Regression	Number of Obs = 747
Group variable: Comid	Number of groups = 126
\mathbb{R}^2 :	Obs per group:
Within $= 0.0754$	min = 5
Between = 0.0002	avg = 5.9
Overall = 0.0002	max = 6
	F(6, 615) = 8.35
$Corr(u_i, Xb) = -0.8623$	Prob > F = 0.000

Table 5. Inputs for the Fixed Effects Model.

4.2.1. Fixed Effect Model: CSR Disclosure Performance and Firm Value

The results in Table 4 above shows that our first alternative hypothesis (H1) for this present study is not supported. This is represented by coefficient (β) of 0.313 and p-value of 0.782. It implies that CSR disclosure performance has a positive, but an insignificant relationship on firm value. Our results suggest that CSR disclosure has a limited effect on firm value. Even so, and with regards to the correlation matrix (Table 3), it shows a weak negative relationship between CSR disclosure performance and firm value of -0.13, which also suggests a weak negative relation between CSR disclosures and firm value. This indicates that CSR disclosure may not necessary lead to increasing firm value. Our findings go to support the line of argument by some researchers who assert that

Sustainability **2018**, *10*, 4518

CSR disclosures are becoming largely regulated and routine and thus lead to providing narrow incremental information beyond other companies' CSR disclosure [29]. Our findings is consistent with Horn et al. [29] research study which found a positive but no significant association between CSR disclosure (as measured by KPMG's 2008, 2011, and 2013 survey data) and firm value (proxy by Tobin's Q) in the context of South Africa. The result of this present study shows that there exist a direct relationship between the extent of CSR disclosure performance and firm value/performance of sampled companies. However, the existence of this direct relationship was insignificant; implying that this positive link may not necessarily lead to significant improvement in firm value/performance. This may be the outcome of the associated cost that may be related to the performance of CSR strategies by firms, which may eventually mitigate the overall returns that may be expected from CSR performance. Our result suggests that CSR disclosure performance has the potential to link firm's future value, which is relevant for stakeholder decision making.

Regarding the influence of the control variables on the firm value in regression Model 1, some interesting results are found. The results show a negative, but significant relationship between firm size and firm value (i.e., β of -1.23; p-value of 0.000 at 1% significance). Prior researchers assert that firm size is a usual predictor of firm value/performance, which is associated with higher costs of monitoring as larger firms are more complex and have more arms' length transactions [117]. Thus, our findings are consistent with prior studies that found a negative association between firm size and firm value/firm performance [118,119]. In the regression results for Model 1, the presence of an external audit firm (i.e., BIG4AUD) is omitted as a result of collinearity, because its value was the same over the period of our analysis. Thus, because there is no variability in its value, this variable was treated as a constant, and that fixed effect transformation eliminate such variables. That is why there were no coefficient values for this variable. The results further show a positive and significant relationship between firm age and firm value (i.e., β of 0.23; p-value of 0.000 at 1% significance). This is consistent with prior research [123] which found that firm age has a significant positive impact on the effects of CSR to firm performance. The results also show a positive and significant relationship between board independence and firm value, (i.e., β of 0.17; p-value of 0.000 at 1% significance). This result suggests that the more independent and nonexecutive directors are added to the board will lead to improve firm value. It also shows a positive, but insignificant relationship between return on asset (profit) and firm value (i.e., β of 0.003; *p*-value of 0.697). Similarly, our result shows a positive, but insignificant correlation between leverage and firm value (i.e., β of 0.012; p-value of 0.729). The result suggests that leverage although is positively related with firm value—it is not significant. This result is consistent with the mixed available evidence on leverage [112].

Tables 6 and 7 provide results for regression model 2 for this study. Model 2 was employed to investigate the relationship among social disclosure performance and environmental disclosure performance, and firm value. In Model 2, the two components along with the control variables are considered as explanatory variables. We checked for multicollinearity problem among these variables through correlation matrix (see Table 3). The result does not indicate any serious collinearity among these variables. Therefore, the two components of CSR disclosure performance are included in one regression model. Therefore, in Table 6, the relevant statistics to take note of are the coefficient (β) of the regressors and p-values. The significance levels are set to 95% with p-values ≤ 0.001 , ≤ 0.05 , and ≤ 0.10 interpreted to be significant.

Sustainability **2018**, 10, 4518 18 of 28

Fixed Effects (within) Regression	Number of Obs = 747		
Group variable: Comid	Number of groups = 126		
\mathbb{R}^2 :	Obs per group:		
Within = 0.0796	$\min = 5$		
Between = 0.0001	avg = 5.9		
Overall = 0.0001	max = 6		
	F(7,614) = 7.59		
Corr $(u_i, Xb) = -0.8605$	Prob > F = 0.0000		

Table 7. Fixed Effects Model 2 (with Social and Environmental Disclosures).

Q	Coef. Std. Err. t $p > t $ [95% Conf. Interval				f. Interval]		
Env_DISC	-1.545	1.093	-1.41	0.158	-3.691	0.600	
Soc_DISC	2.409	1.485	1.62	0.104	-0.506	5.325	
FSIZE	-1.270	0.227	-5.58	0.000	-1.717	-0.823	
BIG4AUD		0 (omitted)					
FAGE	0.231	0.044	5.29	0.000	0.145	0.317	
BODIND	0.165	0.048	3.44	0.001	0.071	0.259	
ROA	0.004	0.009	0.39	0.697	-0.015	0.022	
LEV	0.014	0.034	0.41	0.684	-0.054	0.082	
_cons	20.066	4.315	4.65	0.000	11.593	28.539	
Industry effects		YES					
Sigma_u		8.121					
Sigma_e	1.688						
Rho	0.959 (fraction of variance due to u_i)						
F test that all u_i=0:	F(125, 614) = 32.62 $Prob > F = 0.0000$						

4.2.2. Fixed Effects Model: Environmental Disclosure Performance and Firm Value

In Table 7 the result shows that our second alternative hypothesis (H2) for this present study is also not supported. This is represented by β of -1.545 and p-value of 0.158. It implies that environmental disclosure performance has a negative and insignificant relationship with firm value. Even so, and with regards to the correlation matrix (Table 3), it shows a weak negative relationship between environmental disclosure performance and firm value of -0.12, which also suggests a weak negative relation between environmental disclosures and firm value. Our result is consistent with Verbeeten F.H.M. et al.'s [121] findings of a negative association between environmental disclosure and firm value in a German context. Also, Cormier & Magnan [122] found that environmental reporting does not significantly influence the stock market valuation of Canadian and French firms' earnings. Again, Li et al. [123] found an insignificant relationship between environmental performance and financial performance, and a negative relationship between environmental disclosure and financial performance in a Chinese context. Our results imply that environmental issues could be relatively less prioritised among stakeholder groups compared with social issues in South Africa. This insignificantly negative relationship may be of particular interest to players in the corporate sector of sub-Saharan Africa.

4.2.3. Fixed Effects Model: Social Disclosure Performance and Firm Value

Finally, in Table 6 the result shows that our third alternative hypothesis (H3) for this present study is supported. This is represented by β of 2.41 and p-value of 0.104 (which is significant at 10%). This implies that social disclosure performance has a positive and significant relationship with firm value. This result is consistent with Menassa [124] finding of a positive and significant relationship between social disclosure and firm financial performance (proxy by return on asset and net profit margin).

Sustainability **2018**, 10, 4518

Also, Richardson and Welker [125] found that social disclosures had a positive impact on return on equity. Our result may support the notion that social issues are more relevant to larger stakeholder groups in developing countries, particularly sub-Sahara African, compared with environmental issues, notwithstanding the fact that environmental issues could all connote into social issues from different perspectives. This supports the general assertion that greater attention has been paid towards social issues in the developing countries context by both governmental and corporate institutions; this is evident from various social interventions pursued by various corporate and governmental agencies in such countries. Visser's [126] study of CSR in Africa noted that CSR is mostly associated with philanthropy or charity, i.e., through corporate social investment in education, health, sports, and other community services. It further includes tackling HIV/AIDS, improving working conditions, provision of basic services, supply chain integrity, and poverty alleviation [126].

5. Robustness Test

In this section, we attempt to deal with the problem of endogeneity and to examine the robustness of our findings. In doing this, and consistent with prior studies that used two-stage least squares (2SLS) [127] and one-year lag [112], [127] approach, we use one-year lagged of the three disclosure performance indexes in the robustness test. Thus, we introduced one-year lag between disclosure variables and firm value, such that a firm's performance in any year (Q_t) depends on the previous year's CSR disclosure performance (DISC $_{t-1}$). Hart and Ahuja [128] assert that the benefits from CSR might only become fully visible over a long term, rather than a short term. We therefore argue that disclosure of CSR performance by firms may not have an immediate effect on firms' value/performance, and that it may take time to reflect on firm value/performance. Therefore, we see a one-year lag in the three disclosure indexes. Tables 8 and 9 show the robustness test results. Table 8 shows the CSR disclosure performance index for Model 1; Table 9 shows the environmental and social disclosure performance indexes for Model 2. In all models, we included the control variables to account for factors other than the theoretical constructs of interest that could explain variations in both independent and dependent variables. In Tables 8 and 9, the outcome still remains consistent with the main results for this study.

Table 8. Two-stage least-squares regression (Model 1).

TOBINSQ	Coef. Std.Err. t $p > t $ [95% Conf. Inter-			. Interval]		
lag_CSR_DISC	0.369	2.096	0.18	0.860	-3.747	4.485
FSIZE	-0.423	0.077	-5.47	0.000	-0.576	-0.271
BIG4AUD	-0.033	0.443	-0.07	0.941	-0.903	0.837
FAGE	-0.008	0.006	-1.27	0.204	-0.021	0.004
ROA	-0.004	0.022	-0.19	0.849	-0.048	0.039
LEV	-0.026	0.076	-0.34	0.735	-0.176	0.124
_cons	11.337	1.992	5.69	0.000	7.424	15.249
Industry-effects			Y	ES		
Endogenous variables:			TOB	INSQ		
Exogenous variables:	lag_CSR_DISC FSIZE 0.BIG4AUD 1.BIG4AUD FAGE ROA LEV 1.Sector 2.Sector 3.Sector 4.Sector 5.Sector 6.Sector 7.Sector					
Equation	Obs	Parms	RMSE	"R-sq"	F-Stat	P
TOBINSQ	621	12	4.342	0.127	7.38	0.000

Sustainability **2018**, 10, 4518 20 of 28

TOBINSQ	Coef.	Std.Err.	t	p > t	[95% Conf	. Interval]
lag_Env_DISC	-4.121	2.217	-1.86	0.163	-8.474	0.232
lag_Soc_DISC	3.545	1.653	2.14	0.062	0.299	6.791
FSIZE	-0.412	0.077	-5.33	0.000	-0.564	-0.260
BIG4AUD	-0.009	0.441	-0.02	0.984	-0.875	0.858
FAGE	-0.006	0.007	-0.95	0.344	-0.019	0.007
ROA	-0.004	0.022	-0.18	0.861	-0.047	0.039
LEV	-0.017	0.076	-0.22	0.823	-0.167	0.133
_cons	11.44583	1.985	5.77	0.000	7.547	15.34
Endogenous variables:			TOBIN	ISQ		
Exogenous variables:	lag_Env_DISC lag_Soc_DISC LEV 1.Sector 2.Sector 3.					
Equation	Obs	Parms	RMSE	"R-sq"	F-Stat	P
TOBINSQ	621	13	4.326	0.135	7.30	0.000

Table 9. Two-stage least-squares regression (Model 2).

In the robustness test (Tables 8 and 9) our particular concern is with the main variables of interest for this study. That is, the relationships among the dependent and independent variables. The results as indicated in Table 8 above remain positive and insignificant for Model 1; represented by a β of 0.401 and a p-value of 0.609. This implies that there is a positive, but an insignificant relationship between CSR disclosure performance and firm value. On the other hand, Table 9, Model 2 shows that environmental disclosure performance has a negative and insignificant relationship with firm value; represented by a β of -4.121 and a p-value of 0.163. For the social disclosure performance it shows a positive and a significant relationship with firm value; represented by β of 3.55 and a p-value of 0.062. These results are consistent with our earlier findings for this study. It is particularly worth noting that firm size is the most significant control variable in this study, as its significance (at 1%) is seen in all the regression models, including our robustness test. This is consistent with scholars that have argued that the firm size remains the most significant variable in corporate finance research [107].

6. Summary and Conclusions

Numerous research studies examining the CSR-FP relationship from the perspective of international standardisation, have mainly concentrated on the developed and emerging markets of America, Asia-Pacific, and Europe. This leaves a conspicuous gap within the extant literature with regard to evidence from the sub-Saharan Africa. Nevertheless, some sub-Saharan African countries, particularly South Africa, offer interesting context with regards to investigating CSR disclosure from an international viewpoint. Compared with its peers, South Africa has a matured corporate sector, is ethnically diverse, and has strong regulatory and governance frameworks comparable to those of some developed and other established emerging markets. Therefore this study has attempted to fill the existing gap in the literature from a sub-Saharan African context. This present study has investigated the relationship between CSR disclosure performance (and its components) and firm value. The authors construct three separate disclosure indexes, on the basis of an internationally accepted standard for disclosure of CSR information, to re-examine the questionable CSR-FP relations. Thus, we have investigated the influence of CSR disclosure performance, social disclosure performance and environmental disclosure performance on firm value, in a single study from an emerging economy. We have explored these relations from a multi-theoretical perspective, utilising the stakeholder and legitimacy theories. Hence, this study provides new evidence from an emerging market context with unique institutional setting.

This study supports existing literature which provides evidence that CSR continuous to gain much prominence in South Africa, and for that matter sub-Saharan Africa. Even so, from the descriptive statistics (Table 2) we find that the mean score of CSR disclosure (and the components of CSR) is

Sustainability **2018**, 10, 4518 21 of 28

approximately 60% and above, which gives an indication that on average, the extent of CSR disclosure performance (as measured by the social and environmental disclosure performance indicators on the basis of the G3.1 guidelines) towards sustainability development is moderately high in South Africa. Our findings of a positive but insignificant relationship between CSR disclosure performance and firm value and a negative and an insignificant relationship between environmental disclosure performance and firm value, may suggest a weakening of the influence of CSR disclosure on firm value. From the legitimacy theoretical perspective, we may conclude that CSR disclosure performances and environmental disclosure are avenues by which organisations turn to legitimise their actions and inactions, notwithstanding its usefulness in explaining the motives behind CSR disclosures. From the stakeholder theoretical perspective, we may conclude that disclosures are ways of providing vital nonfinancial information to stakeholder about the firm, which could influence stakeholders' perceptions about the firm. Even so, we are of the view that CSR disclosures, on the basis of international standardisation in developing countries, particularly Sub-Sahara Africa, are largely ways by which firms legitimise their actions and inactions towards powerful stakeholders. Our finding of a positive and statistically significant influence of social disclosure on firm value is evident of many efforts by governments in sub-Saharan Africa and corporate bodies' provision of many social interventions to their communities. For instance, the post-apartheid era saw the formation of many social interventions by the ANC government towards national and corporate rebuilding. Therefore, many different organisations of different sizes and from a variety of different sectors are actively utilising social responsibility policies, not only because of the growing global trends and other external pressures, but also because it could result in efficiency gains. Porter and Kramer [129] indicate that addressing societal concerns amongst others could increase the levels of company productivity, with the subsequent positive effects in profitability and share value.

Therefore, based on our findings the following insights can be drawn. We argue that disclosure of CSR performance, on the basis of the GRI guidelines, serves to provide a common playing field to enable companies to be assessed from an internationally accepted standard. However, this could merely be ceremonial, mainly done for legitimacy purpose. Prior studies [130,131] suggest that companies use several legitimisation strategies to legitimise their actions and inactions. The process of legitimisation can involve real, material change in an organisation's output, methods, goals and disclosure pattern, which in this case, are needed to inform relevant stakeholders about these changes [38, 49]. On the other hand, disclosures can be symbolic, which means that they are part of a legitimisation process whereby companies try to influence the perception of the public without changing performance [132]. By fulfilling this however, it allows disclosing companies to benchmark their performance against others, from a global standard. Benchmarking the disclosure of performance is beneficial in allowing companies to communicate to stakeholders the wellbeing of their companies' operations, as well as, enable their self-assessment [133].

In the wake of global upsurge in international standards and its adoption on debates within the broader framework of corporate sustainable, and the social and environmental disclosure literature, this study makes significant contributions to the existing literature by pushing forward our understanding of the importance of expanding empirical evidence on the CSR–FP relations to different institutional settings that may allow for further findings in light of available inconclusive results within the CSR–FP nexus. We do so following international standards from an emerging institutional setting, where investor protection is relatively low, as a result of a weaker external corporate regulatory environment [12,13].

Although we recognise that our study is not without limitations, nevertheless, our results have important policy implications. It is well recognised that the GRI guidelines is an international and a multi-stakeholder effort to create a common platform for CSR disclosure in line with the economic, environmental, and social impact of organisations. Thus, it seeks to improve the comparability and credibility of sustainability disclosure worldwide. However, GRI guidelines may be far from reach due to the differences in the effects associated with other individual national/institutional settings, culture

Sustainability **2018**, 10, 4518 22 of 28

and social value aspects of markets in developed and developing countries. For instance, research has particularly observed with concerns current CSR approaches and with their origins in developed countries, may not reflect and fully respond to the developing countries' context and circumstances encountered [134]. Further, Jones [135] observed that the national sociocultural environment and level of national economic development are important factors influencing disclosure of CSR and its practice. Thus, international standards for sustainability disclosure, like the GRI, may not fully reflect these observations from different country perspective, and as such rendering such standards relatively less applicable in emerging markets, like sub-Saharan Africa. Thus, we argue that the notion of uniformity for GRI disclosure in line with sustainability, across the length and breadth of organisations globally, may casts serious doubts on its effectiveness to provide a common platform to assess organisations from different institutional contexts. This may lead to impaired outcomes to provide a common platform to assess organisations with similar sectors from different geographical locations. Thus, we find the disclosure parameters lacking specific national and sociocultural value dimensions from different national context, of developed and developing countries. Therefore, we suggest that current regulation and practice in line with international standards (in particular the GRI) should be improved by way of incorporation of other national/institutional outlook and sociocultural value dimensions from the perspective of developed and developing countries to fully reflect the big picture. Therefore, we argue that it is only when these several country level indicators are incorporated that an equitable performance assessment among countries (and within sectorial industries) was fairly possible. Therefore, we assert that the adoption of GRI guidelines for CSR disclosure is inevitably influenced by various institutional locations, regulatory controls, and national sociocultural value dimensions.

Our study has the following limitations, which provide possible future research avenues. First, the process of constructing the disclosure indexes may be bias and potentially subject to measurement error. As such, our results should be interpreted with caution. Even so, we have manually collected the CSR disclosure information from the annual and CSR/sustainability reports of sampled companies and, as such, may result in some omission on our part, although to the best of our knowledge, all relevant information has been captured. Secondly, as we align the CSR information collected with the GRI G3.1 guidelines, to ascertain the extent of disclosure, there could arise elements of subjectivity on our part which could prejudice our results. However, in an attempt to minimise these biases, we acknowledge that every effort were made by the authors to remain objective in the scoring and coding process. As such, all information collected were analysed twice by the authors to ensure that all relevant information are captured and interpreted correctly. Potential research avenues from this paper could be directed at exploring the relationship among the various subcomponents of the individual components of CSR disclosure performance (i.e., environmental and social) and firm value, in order to ascertain further how these other sub-components influence firm value/performance. Also, future research could investigate this relationship within industrial sectors. The future research could also consider other firm value indicators (i.e., both accounting-based and market-based performance, such as ROE and market value) to investigate the relationship between CSR disclosure performance and firm value. Even so, and consistent with prior literature, scholars [69], [82], [136] assert that the CSR-FP relationship depends on various interactions between many different factors, in particular the specific capabilities of the firm and other industrial context. To this end, we postulate that the dynamism that lies within the CSR-FP relations (and its disclosure) research remains inconclusive, owing to the various indirect relations of several indirect variables that have been perceived to influence such relations.

Supplementary Materials: The following are available online at http://www.mdpi.com/2071-1050/10/12/4518/s1, GRI-G3.1 Guidelines for Environmental and Social Performance Indicators.

Author Contributions: This research paper was written as part of a research project. F.S. was responsible for generating the research idea, data collection, coding, performing the analyses, and final write-up of this study. S.N. provided supervisory guidance throughout the whole research process. K.O.B. and K.A.W. assisted with data collection, cleaning, coding, analyses and proof reading.

Sustainability **2018**, 10, 4518 23 of 28

Funding: This research was funded by the Ministry of Education of Humanities and Social Science Grant of China (Grant number: 17YJC790127). The authors gratefully acknowledge the financial support of the Ministry of Education of Humanities and Social Science Grant of China (Grant number: 17YJC790127) to publish in open access.

Conflicts of Interest: The authors declare no conflicts of interest. The sponsors had no role in the design; in the collection of data, analyses, or interpretation of data; in the writing of the manuscript, and in the decision to publish the results of this study.

References

- 1. Baskin, J. Corporate responsibility in emerging markets. J. Corp. Citizsh. 2006, 24, 29–47. [CrossRef]
- 2. Jenkins, H. A business opportunity model of corporate social responsibility for small-and medium-sized enterprises. *Bus. Ethics Eur. Rev.* **2009**, *18*, 21–36. [CrossRef]
- 3. Dunphy, D. Corporate Sustainability: Challenge to managerial orthodoxies. *J. Aust. N. Z. Acad. Manag.* **2003**, 9, 2–11.
- 4. Kuada, J.; Hinson, R.E. Corporate Social Responsibility (CSR) Practices of Foreign and Local Companies in Ghana. *Thunderbird Int. Bus. Rev.* **2012**, *54*, 521–536. [CrossRef]
- 5. Hinson, R.E. "Online CSR reportage of award-winning versus non-award winning banks in Ghana". *J. Inf. Commun. Ethics Soc.* **2011**, *9*, 102–115. [CrossRef]
- 6. Jenkins, H.; Yakovleva, N. Corporate social responsibility in the mining industry: Exploring trends in social and environmental disclosure. *J. Clean. Prod.* **2006**, *14*, 271–284. [CrossRef]
- 7. Marimon, F.; Alonso-Almeida, M.d.M.; Rodríguez, M.d.P.; Alejandro, K.A.C. The world wide diffusion of the global reporting initiative: What is the point? *Clean. Prod.* **2012**, *33*, 132–144. [CrossRef]
- 8. Ligteringen, E.; Zadek, S. The future of corporate responsibility standards. Account. Forum. 2004, 4, 6–17.
- 9. Bernardi, C.; Stark, A.W. Environmental, social and governance disclosure, integrated reporting, and the accuracy of analyst forecasts. *Br. Account. Rev.* **2018**, *50*, 16–31. [CrossRef]
- KPMG. KPMG International Survey of Corporate Responsibility Reporting. 2011. Available online: http://www.kpmg.com/global/en/issuesandinsights/articlespublications/corporate-responsibility/pages/2011-survey.aspx (accessed on 31 January 2018).
- 11. KPMG. KPMG International Survey of Corporate Responsibility Reporting. 2013. Available online: http://www.kpmg.com/global/en/issuesandinsights/articlespublications/corporate-responsibility/pages/2013-survey.aspx (accessed on 31 January 2018).
- 12. LaPorta, R.; Silanes, F.L.; Shleifer, A.; Vishny, R. Investor Protection and Corporate Governance. *J. Financ. Econ.* **2000**, 48–42, 3–27. [CrossRef]
- 13. Ntim, C.; Opong, K.; Danbolt, J. The Relative Value Relevance of Shareholder versus Stakeholder Corporate Governance Disclosure Policy Reforms in South Africa. *Corp. Gov. Int. Rev.* **2012**, *20*, 84–105. [CrossRef]
- 14. Dunning, J.H. (Ed.) *Making Globalization Good: The Moral Challenges of Global Capitalism;* Oxford University Press: Oxford, UK, 2004.
- 15. Dunning, J.H. The Moral Imperatives of Global Capitalism: An Overview. In *Making Globalization Good: The Moral Challenges of Global Capitalism*; Dunning, J.H., Ed.; Oxford University Press: Oxford, UK, 2003.
- 16. Kolk, A. Sustainability, accountability and corporate governance: Exploring multinationals' reporting practices. *Bus. Strateg. Environ.* **2008**, *17*, 1–15. [CrossRef]
- 17. Sotorrio, L.L.; Sanchez, J.L.F. Corporate social reporting for different audiences: The case of multinational corporations in Spain. *Corp. Soc. Responsib. Environ. Manag.* **2010**, 17, 272–283. [CrossRef]
- 18. Gallear, D.; Ghobadian, A.; Chen, W. Corporate responsibility, supply chain partnership and performance: An empirical examination. *Int. J. Prod. Econ.* **2012**, *140*, 83–91. [CrossRef]
- 19. Kim, Y.; Park, M.S.; Wier, B. Is earnings quality associated with corporate social responsibility? *Account. Rev.* **2012**, *87*, 761–796. [CrossRef]
- 20. Hammond, T.; Clayton, B.A.; Arnold, P.J. South Africa's Transition from Apartheid: The Role of Professional Closure in the Experience of Black Chartered Accountants. *Account. Organ. Soc.* **2009**, 34, 705–721. [CrossRef]
- 21. Sampong, F.; Boahene, K.O.; Kweitsu, G. "Apartheid-Induced" Corporate Social Responsibility in the Context of South Africa: A Review. *J. Dev. Ctry. Stud.* **2018**, *8*, 62–71.
- 22. Bai, X.; Chang, J. Corporate social responsibility and firm performance: The mediating role of marketing competence and the moderating role of market environment. *Asia Pac. J. Manag.* **2015**, *32*, 505–530. [CrossRef]

Sustainability **2018**, 10, 4518 24 of 28

23. Labuschagne, C.; Brent, A.C.; van Erck, R.P.G. Assessing the sustainability performances of industries. *J. Clean. Prod.* **2005**, *13*, 373–385. [CrossRef]

- 24. Schadewitz, H.; Niskala, M. Communication via responsibility reporting and its effect on firm value in Finland. *Corp. Soc. Responsib. Environ. Manag.* **2010**, *17*, 96–106. [CrossRef]
- 25. Miralles-quir, M.; Miralles-quir, L.; Miguel, L.; Gon, V. The Value Relevance of Environmental, Social, and Governance Performance: The Brazilian Case. *Sustainability* **2018**, *10*, 574. [CrossRef]
- 26. Mitchell, C.G.; Hill, T. Corporate social and environmental reporting and the impact of internal environmental policy in South Africa. *Corp. Soc. Responsib. Environ. Manag.* **2009**, *16*, 48–60. [CrossRef]
- 27. South Africa. King III Corporate Governance Report for South Africa. 2009. Available online: http://www.iodsa.co.za/?kingII (accessed on 31 January 2018).
- 28. Tilt, C.A. Corporate social responsibility research: The importance of context. *Int. J. Corp. Soc. Responsib.* **2016**, *1*. [CrossRef]
- 29. Horn, R.; de Klerk, M.; de Villiers, C. The association between corporate social responsibility reporting and firm value for South African firms. *S. Afr. J. Econ. Manag. Sci.* **2018**, *21*, 1–10. [CrossRef]
- 30. Chetty, S.; Naidoo, R.; Seetharam, Y. The Impact of Corporate Social Responsibility on Firms' Financial Performance in South Africa. *Contemp. Econ.* **2015**, *9*, 193–214. [CrossRef]
- 31. Ntim, C.G.; Soobaroyen, T. Black Economic Empowerment Disclosures by South African Listed Corporations: The Influence of Ownership and Board Characteristics. *J. Bus. Ethics* **2013**, *116*, 121–138. [CrossRef]
- 32. Cahan, S.F.; van Staden, C.J. Black economic empowerment, legitimacy and the value added statement: Evidence from post-apartheid South Africa. *Account. Financ.* **2009**, *49*, 37–58. [CrossRef]
- 33. Ntim, C.G. Corporate governance, corporate health accounting, and firm value: The case of HIV/AIDS disclosures in Sub-Saharan Africa. *Int. J. Account.* **2016**, *51*, 155–216. [CrossRef]
- 34. Andreason, S. Understanding corporate governance reforms in South Africa: Anglo American divergence, the King reports and hybridization. *Bus. Soc.* **2011**, *50*, 647–673. [CrossRef]
- 35. Statistics South Africa. Census 2011. Statistical Release. P0301.4. 2012. Available online: www.statssa.gov. za/publications/P03014/P030142011.pdf (accessed on 31 January 2018).
- 36. Gyapong, E.; Monem, R.M.; Hu, F. Do Women and Ethnic Minority Directors Influence Firm Value? Evidence from Post-Apartheid South Africa. *J. Bus. Financ. Account.* **2016**, *43*, 370–413. [CrossRef]
- 37. Nyasha, S.; Odhiambo, N.M. The Impact of Banks and Stock Market Development on Economic Growth in South Africa: An ARDL-bounds Testing Approach. *Contemp. Econ.* **2015**, *9*, 193–214. [CrossRef]
- 38. De Villiers, C.; van Staden, C.J. Can less environmental disclosure have a legitimising effect? Evidence from Africa. *Account. Organ. Soc.* **2006**, *31*, 763–781. [CrossRef]
- 39. Suchman, M.C. Managing legitimacy: Strategic and institutional approaches. *Acad. Manag. Rev.* **1995**, *20*, 571–606. [CrossRef]
- 40. Haack, P.; Schoeneborn, D.; Wickert, C. Talking the talk, moral entrapment, creeping commitment? Exploring narrative dynamics in corporate responsibility standardisation. *Organ. Stud.* **2012**, *33*, 815–845. [CrossRef]
- 41. Perez-Batres, L.; Doh, J.; Miller, V.; Pisani, M. Stakeholder pressures as determinants of CSR strategic choice: Why do firms choose symbolic versus substantive self-regulatory codes of conduct? *J. Bus. Ethics* **2012**, *110*, 157–172. [CrossRef]
- 42. Slager, R.; Gond, J.P.; Moon, J. Standardization as institutional work: The regulatory power of a responsible investment standard. *Organ. Stud.* **2012**, *33*, 763–790. [CrossRef]
- 43. Moir, L. What do we mean by corporate social responsibility? Corp. Gov. 2001, 1, 16–22. [CrossRef]
- 44. Garriga, E.; Melé, D. Corporate Social Responsibility Theories: Mapping the Territory. *J. Bus. Ethics* **2004**, *53*, 51–71. [CrossRef]
- 45. Jamali, D.; Mirshak, R. Corporate social responsibility (CSR): Theory and practice in a developing country context. *J. Bus. Ethics* **2007**, 72, 243–262. [CrossRef]
- 46. Palazzo, G.; Scherer, A.G. Corporate legitimacy as deliberation: A communicative framework. *J. Bus. Ethics* **2006**, *66*, 71–88. [CrossRef]
- 47. Branco, M.C.; Rodrigues, L.L. Factors influencing social responsibility disclosure by Portuguese companies. *J. Bus. Ethics* **2008**, *83*, 685–701. [CrossRef]
- 48. Reverte, C. Determinants of corporate social responsibility disclosure ratings by Spanish listed firms. *J. Bus. Ethics* **2009**, *88*, 351–366. [CrossRef]

Sustainability **2018**, 10, 4518 25 of 28

49. Deegan, C. The legitimising effect of social and environmental disclosures—A theoretical foundation. *Account. Audit. Account. J.* **2002**, *15*, 282–311. [CrossRef]

- 50. Freeman, E. Strategic Management: A Stakeholder Approach; Pitman Publishing: Boston, MA, USA, 1984.
- 51. Donaldson, T.; Preston, L.E. The stakeholder theory of the corporation: Concepts, evidence, and implications. *Acad. Manag. Rev.* **1995**, *20*, 65–91. [CrossRef]
- 52. Porter, M.E.; Kramer, M.R. Strategy & society: The link between competitive advantage and corporate social responsibility. *Harv. Bus. Rev.* **2006**, *84*, 78–92. [PubMed]
- 53. Turban, D.B.; Greening, D.W. Corporate Social Performance and Organizational Attractiveness to Prospective Employees. *Acad. Manag.* **1997**, *40*, 656–672.
- 54. Orlitzky, M.; Schmidt, F.L.; Rynes, S.L. Corporate social and financial performance: A meta-analysis. *Organ. Stud.* **2003**, 24, 403–441. [CrossRef]
- 55. Bowman, E.H.; Haire, M. A Strategic Posture toward Corporate Social Responsibility. *Calif. Manag. Rev.* **1975**, *XVIII*, 49–58. [CrossRef]
- 56. Porter, M.E.; Linde, C.V.D. Toward a new conception of the environment-competitiveness relationship. *J. Econ. Perspect.* **1995**, *9*, 97–118. [CrossRef]
- 57. Cornell, B.; Shapiro, A. Corporate Stakeholders and Corporate Finance. *Financ. Manag. J.* **1987**, *16*, 5–14. [CrossRef]
- 58. Berry, M.A.; Rondinelli, D.A. Proactive corporate environmental management: A new industrial revolution. *Acad. Manag. Exec.* **1998**, *12*, 1–13. [CrossRef]
- 59. Dawkins, C.E. First to market: Issue management pacesetters and the pharmaceutical industry response to AIDS in Africa. *Bus. Soc.* **2005**, *44*, 244–282. [CrossRef]
- 60. Cormier, D.; Gordon, I.M.; Magnan, M. Corporate environmental disclosure: Contrasting management's perceptions with reality. *J. Bus. Ethics* **2004**, *49*, 143–165. [CrossRef]
- 61. Woodward, D.G.; Edwards, P.; Birkin, F.K. Organizational Legitimacy and Stakeholder Information Provision. *Br. J. Manag.* **1996**, *7*, 329–347. [CrossRef]
- 62. Mahoney, L.S.; Thorne, L.; Cecil, L.; LaGore, W. A Research Note on Standalone Corporate Social Responsibility Reports: Signaling or Greenwashing? *Crit. Perspect. Account.* **2013**, 24, 350–359. [CrossRef]
- 63. Joseph, G. Ambiguous but tethered: An accounting basis for sustainability reporting. *Crit. Perspect. Account.* **2012**, 23, 93–106. [CrossRef]
- 64. Moneva, J.M.; Archel, P.; Correa, C. GRI and the camouflaging of corporate unsustainability. *Account. Forum* **2006**, *30*, 121–137. [CrossRef]
- 65. De Villiers, C.J.; Alexander, D. The institutionalisation of corporate social responsibility reporting. *Br. Account. Rev.* **2014**, *46*, 198–212. [CrossRef]
- 66. Adams, C.A. The ethical, social and environmental reporting-performance portrayal gap. *Account. Audit. Account. J.* **2004**, *17*, 731–757. [CrossRef]
- 67. Adams, C.A. A commentary on: Corporate social responsibility reporting and reputation risk management. *Account. Audit. Account. J.* **2008**, *21*, 365–370. [CrossRef]
- 68. Margolis, J.D.; Elfenbein, H.A.; Walsh, J.P. Does It Pay to Be Good . . . and Does It Matter? A Meta-Analysis of the Relationship between Corporate Social and Financial Performance. Soc. Sci. Res. Netw. 2009. [CrossRef]
- 69. Baird, P.L.; Geylani, P.C.; Roberts, J.A. Corporate Social and Financial Performance Re-Examined: Industry Effects in a Linear Mixed Model Analysis. *J. Bus. Ethics* **2012**, *109*, 367–388. [CrossRef]
- 70. Saeidi, S.P.; Sofian, S.; Saeidi, P.; Saaeidi, S.A. How Does Corporate Social Responsibility Contribute to Firm Financial Performance? The Mediating Role of Competitive Advantage, Reputation, and Customer Satisfaction. *J. Bus. Res.* **2015**, *68*, 341–350. [CrossRef]
- 71. Aupperle, K.E.; Carroll, A.B.; Hatfield, J.D. An empirical examination of the relationship between corporate social responsibility and profitability. *Acad. Manag. J.* **1985**, *28*, 446–463.
- 72. Vance, S.C. Are socially responsible corporations good investment risks? *Manag. Rev.* 1975, 64, 18–24.
- 73. Alexander, G.J.; Buchholz, R.A. Corporate social responsibility and stock market performance. *Acad. Manag. J.* **1978**, 21, 479–486. [CrossRef]
- 74. Li, F. Endogeneity in CEO power: A survey and experiment. Invest. Anal. J. 2016, 45, 149–162. [CrossRef]
- 75. Cormier, D.; Magnan, M. Environmental reporting management: A continental European perspective. *J. Account. Public Policy* **2003**, 22, 43–62. [CrossRef]

Sustainability **2018**, 10, 4518 26 of 28

76. Cormier, D.; Magnan, M.; van Velthoven, B. Environmental disclosure quality in large German companies: Economic incentives, public pressures or institutional conditions? *Eur. Account. Rev.* **2005**, *14*, 3–39. [CrossRef]

- 77. Harte, G.; Owen, D. Environmental disclosure in the annual reports of British companies: A research note. *Account. Audit. Account. J.* **1991**, *4*, 51–61. [CrossRef]
- 78. Carroll, A.B. A Three-Dimensional Conceptual Model of Corporate Performance. *Acad. Manag. Rev.* **1979**, *4*, 497–505. [CrossRef]
- 79. WCED. Report of the World Commission on Environment and Development: Our Common Future; Oxford University Press: Oxford, UK, 1987.
- 80. Daub, C.; Ergenzinger, R. Enabling sustainable management through a new multi-disciplinary concept of customer satisfaction. *Eur. J. Mark.* **2005**, *39*, 998–1012. [CrossRef]
- 81. Werther, W.B.; Chandler, D. *Strategic Corporate Social Responsibility: Stakeholders in a Global Environment*, 2nd ed.; Sage Publications Inc.: Thousand Oaks, CA, USA, 2010.
- 82. Waddock, S.A.; Graves, S.M. The corporate social performance—Financial performance link. *Strateg. Manag. J.* **1997**, *18*, 303–319. [CrossRef]
- 83. Cochran, P.L.; Wood, R.A. Corporate Social Responsibility and Financial Performance. *Acad. Manag.* **1984**, 27, 42–56.
- 84. Freeman, R.E. Strategic management: A stakeholder approach. Adv. Strateg. Manag. 1983, 1, 31–60.
- 85. Jones, T. Instrumental stakeholder theory: A synthesis of ethics and economics. *Acad. Manag. Rev.* **1995**, *20*, 404–437. [CrossRef]
- 86. Van Beurden, P.; Gössling, T. The Worth of Values—A Literature Review on the Relation between Corporate Social and Financial Performance. *J. Bus. Ethics* **2008**, *82*, 407. [CrossRef]
- 87. Friedman, M. A Friedman doctrine: The social responsibility of business is to increase its profits. *New York Times Magazine*, 13 September 1970.
- 88. Chen, L.; Feldmann, A.; Tang, O. The relationship between disclosures of corporate social performance and financial performance: Evidences from GRI reports in manufacturing industry. *Int. J. Prod. Econ.* **2015**, 170, 445–456. [CrossRef]
- 89. McWilliams, A.; Siegel, D. Corporate social responsibility: A theory of the firm's perspective. *Acad. Manag. Rev.* **2001**, *26*, 117–127. [CrossRef]
- 90. Preston, L.E.; O'Bannon, D.P. The corporate social—Financial performance relationship: A typology and analysis. *Bus. Soc.* **1997**, *36*, 419–429. [CrossRef]
- 91. Laskar, N.; Maji, S.G. Disclosure of Corporate Social Responsibility and Firm Performance: Evidence from India. *Asia Pac. J. Manag. Res. Innov.* **2017**, *12*, 145–154. [CrossRef]
- 92. Carrot and Sticks. Sustainability Reporting Policies Worldwide—Today's Best Practice, Tomorrow's Trends. 2013. Available online: https://www.globalreporting.org/resourcelibrary/Carrots-and-Sticks.pdf (accessed on 30 June 2018).
- 93. Mahlotra, N.K. Marketing RESEARCH: An Applied Orientation; Pearson: London, UK, 2007.
- 94. Patten, D.M. Give or take on the internet: An examination of the disclosure practices of insurance firm web innovators. *J. Bus. Ethics* **2002**, *36*, 247–259. [CrossRef]
- 95. Muttakin, M.B.; Khan, A. Determinants of corporate social disclosure: Empirical evidence from Bangladesh. *Adv. Account.* **2014**, *30*, 168–175. [CrossRef]
- 96. Chen, S.; Bouvain, P. Is corporate responsibility converging? A comparison of corporate responsibility reporting in the USA, UK, Australia and Germany. *J. Bus. Ethics* **2009**, *87*, 299–317. [CrossRef]
- 97. Tate, W.L.; Ellram, L.M.; Kirchoff, J.F. Corporate Social Responsibility Reports: A Thematic Analysis Related to Supply Chain Management. *J. Supply Chain Manag.* **2010**, *46*, 19–44. [CrossRef]
- 98. Chen, L.; Tang, O.; Feldmann, A. Applying GRI reports for the investigation of environmental management practices and company performance in Sweden, China and India. *J. Clean. Prod.* **2015**, *98*, 36–46. [CrossRef]
- 99. Jiao, Y. Corporate disclosure, market valuation, and firm performance. *Financ. Manag.* **2011**, 40, 647–676. [CrossRef]
- 100. Cahan, S.F.; de Villiers, C.; Jeter, D.C.; Naiker, V.; van Staden, C.J. Are CSR disclosures value relevant? Cross-country evidence. *Eur. Account. Rev.* **2016**, *25*, 579–611. [CrossRef]
- 101. De Villiers, C.; Marques, A. Corporate social responsibility, country-level predispositions, and the consequences of choosing a level of disclosure. *Account. Bus. Res.* **2016**, *46*, 167–195. [CrossRef]

Sustainability **2018**, 10, 4518 27 of 28

102. Bhagat, S.; Black, B. The Non-Correlation between Board Independence and Long-Term Firm Performance. *J. Corp. Law* **2002**, 27, 231–274. [CrossRef]

- 103. Thomas, D.E.; Eden, L. What is the Shape of the Multinationality-Performance Relationship? *Multinatl. Bus. Rev.* **2004**, *12*, 89–110. [CrossRef]
- 104. Leightner, J.E.; Inoue, T. Tackling the omitted variables problem without the strong assumptions of proxies. *Eur. J. Oper. Res.* **2007**, *178*, 819–840. [CrossRef]
- 105. Larcker, D.F.; Richardson, S.A.; Tuna, I. Corporate governance, accounting outcomes, and organizational performance. *Account. Rev.* **2007**, *82*, 963–1008. [CrossRef]
- 106. Satish, K.; Thompson, S.H.T. Moderating Effects of Governance on Information Infrastructure and E-Government Development. J. Am. Soc. Inf. Sci. Technol. 2012, 63, 1929–1946.
- 107. Dang, C.; Li, Z.; Yang, C. Measuring firm size in empirical corporate finance. *J. Bank. Financ.* **2018**, *86*, 159–176. [CrossRef]
- 108. Kumar, K.B.; Rajan, R.G.; Zingales, L. *What Determines Firm Size?* (*No. w7208*); National Bureau of Economic Research: Cambridge, MA, USA, 1999.
- 109. Hart, P.E.; Oulton, N. Growth and Size of Firms. Econ. J. 1996, 106, 1242–1252. [CrossRef]
- 110. Guthrie, J.; Johanson, U.; Bukh, P.N.; Sanchez, P. Intangibles and the transparent enterprise: New strands of knowledge. *J. Intellect. Cap.* **2003**, *4*, 429–440. [CrossRef]
- 111. Potter, W.J.; Levine-Donnerstein, D. Rethinking validity and reliability in content analysis. *J. Appl. Commun. Res.* **1999**, 27, 258–284. [CrossRef]
- 112. Ntim, C.G. An Integrated Corporate Governance Framework and Financial Performance in South African-Listed Corporations. *South Afr. J. Econ.* **2013**, *81*, 373–392. [CrossRef]
- 113. Gujarati, D.N. Basic Econometrics, 5th ed.; McGraw-Hill: New York, NY, USA, 2009.
- 114. Schroeder, M.A. Diagnosing and Dealing with Multicollinearity. West. J. Nurs. Res. 1990, 12, 175–187. [CrossRef] [PubMed]
- 115. Greene, W.H. Econometric Analysis, 6th ed.; Prentice-Hall: Upper Saddle River, NJ, USA, 2008.
- 116. Dzingai, I.; Fakoya, M.B. Effect of Corporate Governance Structure on the Financial Performance of Johannesburg Stock Exchange (JSE)-Listed Mining Firms. *Sustainability* **2017**, *9*, 867. [CrossRef]
- 117. Isidro, H.; Sobral, M. The Effects of Women on Corporate Boards on Firm Value, Financial Performance, and Ethical and Social Compliance. *J. Bus. Ethics* **2015**, *132*, 1–19. [CrossRef]
- 118. Carter, D.; D'Souza, F.; Simkins, B.; Simpson, W. The gender and ethnic diversity of US boards and board committees and firm financial performance. *Corp. Gov. Int. Rev.* **2010**, *18*, 396–414. [CrossRef]
- 119. Bohren, O.; Strom, R. Governance and politics: Regulating independence and diversity in the board room. *J. Bus. Financ. Account.* **2010**, *37*, 1281–1308. [CrossRef]
- 120. Moore, G. Corporate Social and Financial Performance: An Investigation in the U.K. Supermarket Industry. *J. Bus. Ethics.* **2001**, *34*, 299–315. [CrossRef]
- 121. Verbeeten, F.H.M.; Gamerschlag, R.; Möller, K. Are CSR disclosures relevant for investors? Empirical evidence from Germany. *Manag. Decis.* **2016**, *54*, 1359–1382. [CrossRef]
- 122. Cormier, D.; Magnan, M. The revisited contribution of environmental reporting to investors' valuation of a firm's earnings: An international perspective. *Ecol. Econ.* **2007**, *62*, *613–626*. [CrossRef]
- 123. Li, D.; Zhao, Y.; Sun, Y.; Yin, D. Corporate environmental performance, environmental information disclosure, and financial performance: Evidence from China. *Hum. Ecol. Risk Assess. Int. J.* **2017**, 23, 323–339. [CrossRef]
- 124. Menassa, E. Corporate social responsibility: An exploratory study of the quality and extent of social disclosures by Lebanese commercial banks. *J. Appl. Account. Res.* **2010**, *11*, 4–23. [CrossRef]
- 125. Richardson, A.J.; Welker, M. Social disclosure, financial disclosure and the cost of equity capital'. *Account. Organ. Soc.* **2001**, *26*, 597–616. [CrossRef]
- 126. Visser, W. Revisiting Carroll's CSR pyramid. In *Corporate Citizenship in Developing Countries: New Partnership Perspectives*; Copenhagen Business School Press: New York, NY, USA, 2006.
- 127. Sial, M.S.; Zheng, C.; Khuong, N.V.; Khan, T.; Usman, M. Does firm performance influence corporate social responsibility reporting of Chinese listed companies? *Sustainability* **2018**, *10*, 2217. [CrossRef]
- 128. Hart, S.; Ahuja, G. Does it pay to be green? An empirical examination of the relationship between emission reduction and firm performance. *Bus. Strateg. Environ.* **1996**, *5*, 30–37. [CrossRef]
- 129. Porter, M.E.; Kramer, M.R. Creating shared value. How to reinvent capitalism-and unleash a wave of innovation and growth. *Harv. Bus. Rev.* **2011**, *89*, 62–77.

Sustainability **2018**, 10, 4518 28 of 28

130. Dowling, J.; Pfeffer, J. Organizational legitimacy: Social values and organisational behaviour. *Pac. Assoc. Rev.* **1995**, *18*, 122–136.

- 131. Lindblom, C.K. The implications of organizational legitimacy for corporate social performance and disclosure. In Proceedings of the Critical Perspectives on Accounting Conference, New York, NY, USA, 13–15 June 1994.
- 132. Neu, D.; Warsame, H.; Pedwell, K. Managing public impressions: Environmental disclosures in annual reports. *Account. Organ. Soc.* **1998**, 23, 265–282. [CrossRef]
- 133. Skouloudis, A.; Evangelinos, K.; Kourmousis, F. Assessing non-financial reports according to the Global Reporting Initiative guidelines: Evidence from Greece. *J. Clean. Prod.* **2010**, *18*, 426–438. [CrossRef]
- 134. Muthuri, J.N.; Gilbert, V. An Institutional Analysis of Corporate Social Responsibility in Kenya. *J. Bus. Ethics* **2011**, *98*, 467–483. [CrossRef]
- 135. Jones, M. The Institutional Determinants of Social Responsibility. J. Bus. Ethics 1999, 20, 163–179. [CrossRef]
- 136. Griffin, J.F.; Mahon, J.J. The corporate social performance and corporate financial performance debate: Twenty-five years of incomparable research. *Bus. Soc.* **1997**, *36*, 5–31. [CrossRef]



© 2018 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).